Author's response to reviews

Title: Promoting recovery-oriented practice in mental health services: a quasi-experimental mixed-methods study

Authors:

Helen Gilburt (helen.gilburt@kcl.ac.uk)
Mike Slade (mike.slade@kcl.ac.uk)
Victoria Bird (victoria.bird@kcl.ac.uk)
Sheri Oduola (sherifat.oduola@kcl.ac.uk)
Tom Craig (thomas.craig@kcl.ac.uk)

Version: 3 Date: 6 June 2013

Author's response to reviews: see over
Dear Prof Emmelkamp,

Many thanks for your prompt response and the reviewer’s comments. As requested, we have dealt with all of the comments that the second reviewer raised.

Reviewer 2: Tom Trauer

1. Spelling rectified as suggested

2. Sentence redrafted to: ‘There was a move from a focus on maintenance towards improvement, looking forward beyond crisis management to what happens next, including when patients left their care and potentially the care of the mental health services altogether.’

3. Table 2 has been amended as suggested and ordered according to frequency of occurrence.

4. Table 3 has been amended as suggested, following the same order as Table 2.

We hope that this addresses the reviewer’s comments sufficiently and we look forward to receiving your response.

Yours Sincerely

Helen Gilburt